Author's response to reviews

Title: People with diagnosed HIV infection not attending for specialist clinical care: UK national review

Authors:

Hilary Curtis (hilary@regordane.net)
Zheng Yin (Zheng.Yin@phe.gov.uk)
Katherine Clay (kate.wheatley@doctors.org.uk)
Alison E Brown (kate.wheatley@doctors.org.uk)
Valerie C Delpech (Valerie.Delpech@phe.gov.uk)
Edmund LC Ong (edmung.ong@newcastle.ac.uk)

Version: 2 Date: 20 November 2014

Author's response to reviews:

This manuscript has been revised in response to a request from the editors, as follows:

Title page has been re-formatted to include author affiliations and contact details.

A statement regarding ethics and informed consent has been added to the Methods section. In this context new reference 10 has been added and previous reference 10 has been re-numbered 11.

Line numbering has been added.

A list of abbreviations has been added.

A statement has been added that the authors have no competing interests.

A description of Authors' contributions has been added.

The Acknowledgements section has been moved to before the References list.

On behalf of the authors I request consideration of this revised manuscript and look forward to hearing from you further.

Hilary Curtis